Author's response to reviews

Title: In-vivo visualisation of the anatomical structures related to the acupuncture points Dai mai and Shen mai by MRI. A single-case pilot study.

Authors:

Roy Moncayo (anmeldung@womed.at)
Ansgar Rudisch (a.rudisch@uibk.ac.at)
Christian Kremser (christian.kremser@uibk.ac.at)

Version: 6 Date: 12 October 2006

Author's response to reviews:

Dear Editorial Office!

The present version includes a 3D-volume rendered reconstruction of the abdominal MRI image. Using this procedure we hope that Dr. Langevin will be able to recognize the muscular structures. The image automatically does color coding, so that muscle is shown in muscle-red. The image was reconstructed using dedicated software from Nuclear Diagnostics, Sweden.